Author's response to reviews

Title: Discrepancy between Effects of Carbapenems and Flomoxef in Treating Nosocomial Hemodialysis Access-related Bacteremia Secondary to Extended Spectrum Beta-lactamase Producing Klebsiella pneumoniae in Patients on Maintenance Hemodialysis

Authors:

Chih-Chao Yang (papaofison@gmail.com)
Shau-Hsuan Li (ma4844@adm.cgmh.org.tw)
Feng-Rong Chuang (ma1805@adm.cgmh.org.tw)
Chih-Hsiung Lee (ma4228@adm.cgmh.org.tw)
Jin-Bor Chen (ma1464@adm.cgmh.org.tw)
Chien-Te Lee (noteofnephrology@gmail.com)

Version: 2 Date: 3 August 2011

Author's response to reviews:

The manuscript has been updated with the information regarding the issues below:

1) an Authors' Contributions section at the end of the manuscript, before the reference list.
2) a 'Competing interests' section after the Conclusions.
3) an acknowledgement section at the end of the manuscript before the reference list.
4) clarify what ethical approval was obtained for your study and add a statement regarding ethics to the methods section(including the name of the body which gave approval, with a reference number where appropriate)